Author’s response to reviews

Title: Availability of Essential Medicines and Pharmaceutical Inventory Management Practice at health centers of Adama town, Ethiopia

Authors:

Adane Kefale (adanet2011@gmail.com)

Hafiza Shebo (hafizahayredin@yahoo.com)

Version: 3 Date: 20 Mar 2019

Author’s response to reviews:

Reviewer’s comment: better to shorten the manuscript

Author’s response: We thank you the reviewer for your valuable comments and concern. We gone through the entire document to minimize the size of the text and tried our best. We rewrite some statements to clarify the idea and shorten it. Because the paper contains two topics (availability of EMs and inventory management practice) it is a bit challenging to further shrink the paper while achieving the objectives of the study. The change we made is presented in track change form in the manuscript.